MOSS 2 Test Readiness Review Minutes

July 8, 2003

ELEMENT ATTENDANCE:

ESDIS SSDO, ESDIS SOO, MRTT, EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, Averstar

NEXT TELECON: MOSS 2 Debrief, August 5, 2003 2pm EST

MEETING HIGHLIGHTS:

All TRR presentations can be obtained from the Aura Testing website: http://jupiter02.gsfc.nasa.gov/AuraTesting/ under MOSS2 link. Only highlights, changes, or clarifications will be presented below.

SSDO:

Karen reviewed the MOSS 2 schedule highlighting:

· Daily dial in telecon: Mon-Thur; 7/14-7/17; 4pm EST; Dial in # 888-395-8106; Passcode 87751

· Requirements status complete July 29

· Debrief – August 5

· Interim SORR – August 12. John Moses stated that this will be a telecon and will involve EDOS, EMOS, FDS, DAACs, and SIPS. John will send out an email with more information soon.

Karen mentioned that requirements for APIDs, which there would be no data in MOSS 2, could be verified through spacecraft data flows. A new Test ID “SC” was added to the requirements database to denote those requirements verified through a spacecraft data flow.

MOSS 3 and MOSS 4 scheduling information are appended to the higher-level Aura Test Schedule posted on the website. MOSS 3 new data are due to ESDIS by August 15. Karen asked all elements to review the Science and Mission System Milestones sections of the Aura Test Schedule and provide any updates to Karen.

MRTT:

Rhonda presented the MRTT slides and mentioned that they are piggy backing on this MOSS test for their JEM-1 test. They plan to verify 13 objectives. It was pointed out by EDOS that the EDOS-HIRDLS UK interface is not ready and is planned for MOSS 3. Rhonda asked the elements to give them a heads up if any deviations are anticipated for a particular test. 

FDS:

FDS provided their slides prior to the TRR. Their interface with GES DAAC is still not ready and they provided the products to the DAAC on a CD. The GES DAAC ingested the files and ran DPREP.

EDOS:

Ron reported that EDOS would participate in MOSS 2 from their System B, the operational system. EDOS is green for all their MOSS 2 requirements except for the requirement for automatic secure ftp to HIRDLS NRTP. The delivery will be performed manually. The secure ftp patch, C4.5b, is not yet installed on the operational system. Ed mentioned that GES DAAC would now run their front end of the system from OPS mode. This results in a re-configuration of EDOS. Ron indicated that a waiver would be required to make the change. Informal testing between GES DAAC and EDOS to occur prior to MOSS 2 formal run.

Ron mentioned that the RBD and VCD files will follow the end of the SCS event and the PDS files will contain a one-hour wait period prior to delivering. For the specific data products, times, and amounts, refer to Bill Watt’s MOSS 2 timeline spreadsheet. He also mentioned that the users should see data about 1 ½ - 2 hours after EDOS has been primed. Ron asked if GBAD APIDs 967, 958, and 959 be delivered to HIRDLS NRTP since the UK interface will not be ready. HIRDLS replied NO.

EMOS:

Utah presented the slide via telecon. The EMOS database version will be PMDB #1. One requirement (619) will be verified during MOSS 2 and three MOSS 2 requirements (601, 620, and 621) will be validated with S/C tests. Thirty-seven GBAD carryout files expected to be processed and delivered to GES DAAC. Ed from the GES DAAC mentioned to EMOS that their interface also would need to point to OPS mode. An interface test is planned for later this week. Utah asked if the GBAD data still have the 8th packet problem. Karen responded that up to day 107 it do but after day 107 (new data) it do not. No problem for EMOS.

GES DAAC:

Ed reported that as a result of the MOSS 2 Dry Run efforts, the GES DAAC would participate in the MOSS 2 Formal in a hybrid configuration using OPS mode for the “Front End” and TS2 for the “Back End”. (See Ed’s TRR slide for the details.) The distribution of OMI L1B products to ODPS in TS2 may not be automatic. Ed mentioned that the GES DAAC would treat this test as an operational exercise for the operations staff. If problems occur, contact the Operation Controller first. If that’s not working contact Ed. Dave Cuddy asked Ed how would MTMGW work at the DAAC? Ed responded that for TS2 mode higher-level products and L0 data from old tests can be searched and ordered. MLS also stated that if it turns out that they cannot produce higher-level products, they would not stage any products; it would be no value added.

OMI SIPS:

Marghi reported that OMI SIPS plans to process the L0 data from GES DAAC and produce L1 standard products as well as some L2 products. L2 production was planned for MOSS 3. As a deviation, simulated L1B data will be used to generate the L2 products. Marghi mentioned that requirements 1732, 1733, and 1734 (zoomed radiance and irradiance products) might not be able to be tested if the spacecraft was not in the right configuration when the data were produced. She also mentioned that some files will fail ingest by the GES DAAC because they have not yet received updates to the PGEs needed to produce the correct .met file.  It was pointed out that the L0 products (OML0A1 and OML0A2 – PDSs) are still being produced by EDOS and sent to GES DAAC for delivery to OSIPS even though the initial requirement were deleted. Ed McDonald requested that OMIS SIPS contact the LP DAAC (EDC) directly to receive the MYD09A1 product. Marghi replied that they would do that and write up an Operations Agreement (OA) with LP DAAC. Marghi reported that April 18th was the last set of files they received for LeapSecT and UTCPoleT. Ed checked on the issue during the telecon and found that the subscription needs to be updated to reference the updated version numbers. GES DAAC user services will take care of it. MTMGW requests will be issued to TS2 mode for higher-level products. Marghi stated that after MOSS 2 they will review their ancillary data list again and make any subscription request changes.

HIRDLS:

Vince presented the slides via telecon. He reported that they will not verify requirement 1329 (QA) in MOSS 2, but will exercise MTMGW. Vince pointed out that the requirements for APID 1630 were not in the MOSS requirements database. After the telecon, Karen added the requirements. Vince mentioned that the Level 2 products would be generated using a staged “HIRRAD” intermediate L2 product.

MLS:

Audrey presented the slides via telecon. She reported that there is no data for APID 1748 in the current MOSS 2 data for MLS. Dave reported that ESDIS should have a small amount of this data. After the telecon, Eric located the data (4 packets). In concurrence with Dave, it was decided to not time shifted that data and play it back as part of MOSS 2 at the end of the data flows.

MLS mentioned that they have been receiving LeapSecT and UTCPoleT files from the GES DAAC as well as obtaining them via a toolkit call. MLS raised the issue of the Toolkit 5.2.9.1 error causes their processing to stop; the program will continue if a warning is issued. After the telecon, a toolkit patch was released to MLS to correct the problem. Also there was a concern about large gaps being in the MOSS 2 L0 data. Karen replied that the gaps were due to the MOSS 2 Dry Run and for the MOSS 2 Formal the data would be contiguous.

ASDC DAAC:

Sue presented the slides via telecon and reported that ASDC DAAC is ready to verify the 101 MOSS 2 requirements and regression test the MOSS 1 requirements. ASDC pointed out that the MOSS 2 Dry Run did not contain L0 data for every TES expedited data type and asked should requirements be failed for any missing data type not received from EDOS during MOSS 2. The answer was: the requirement should be status as “not tested”.

TES SIPS:

Andy presented the slides via telecon and reported that 50 MOSS 2 requirements are allocated to TES SIPS. MTMGW will be used to re-order L0 data. TES mentioned that they also receive DAO and DPREP data via subscription from the ASDC DAAC. The TES software will not be using DPREP data to produce any products.

Averstar:

Scott presented the slides via telecon. Scott mentioned that data would not be re-ordered if successful in previous MOSS runs. Karen asked Scott to verify the additional OMI products identified in Marghi’s presentation and order if available.

ODPS held in separate telecon July 8th 10am EST

ACTION ITEMS: Documented in the Aura Action Item Tool at http://jupiter02.gsfc.nasa.gov/auratesting/ .  Please remember to enter your status in the tool.
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